\documentstyle[12pt]{report}

\pagestyle{myheadings}
\def\headlineaux{\sc Math 101: Algebra for College Students
\hrulefill
}
\markboth{\headlineaux}{\headlineaux}

\setlength{\textwidth}{6.25in}
\setlength{\oddsidemargin}{0in}
\setlength{\evensidemargin}{0in}

\begin{document}

\begin{titlepage}
Blank Page
\end{titlepage}

\begin{minipage}{2.5in}
Prof. Burt Rosenberg \\
Ungar 523, x--2575
\end{minipage}
\hfill
\begin{minipage}{3in}
\begin{flushright}
MTH 101--P Term 964\\
Time: M--F 8.30--9.55 A.M.\\
Room: MM 107
\end{flushright}
\end{minipage}

\begin{itemize}

\item {\bf Text:}
{\sc Algebra for College Students} by Larson and Hosteltler.

\item {\bf Tests and Grading:}\\
Grading will be 40\% on submitted homeworks, 60\% on the
five exams. There are no make-up exams.

\item {\bf Syllabus and Homework Assignments:}\\
Homework assignments are due on the date given below.
Late homeworks are not accepted.
\begin{itemize}
\item{\bf Week 1.}
   \begin{itemize}
   \item{\it Wed., June 26:}
      Ch 1. Introduction to Algebra:
      Sections 1.1 and 1.2.
   \item{\it Thr., June 27:}
      Sections 1.3 and 1.4.
   \item{\it Fri., June 28:}
      Section 1.5.
      Ch 2. Linear Equations and Inequalities:
      Section 2.1.
      {\bf Homework due:} Odd numbered problems from 1.1--1.4.
   \end{itemize}
\item{\bf Week 2.}
   \begin{itemize}
   \item{\it Mon., July 1:}
      Sections 2.2 and 2.3.
   \item{\it Tue., July 2:}
      Sections 2.4 and 2.5.
      {\bf Homework due:} Odd numbered problems from 1.5 and 2.1--2.3.
   \item{\it Wed., July 3:}
      {\bf Exam 1: Chapters 1 and 2.}
   \item{\it Thr., July 4: Vacation.}
   \item{\it Fri., July 5:}
      Ch 3. Polynomials and Factoring:
      Sections 3.1 and 3.2.
   \end{itemize}
\item{\bf Week 3.}
   \begin{itemize}
   \item{\it Mon., July 8:}
      Sections 3.3 and 3.4.
   \item{\it Tue., July 9:}
      Section 3.5 and review.
      {\bf Homework due:} Odd numbered problems from 3.1--3.4.
   \item{\it Wed., July 10:}
      {\bf Exam 2: Chapter 3.}
   \item{\it Thr., July 11:}
      Ch 4. Rational Expressions: Sections 4.1 and 4.2.
   \item{\it Fri., July 12:}
      Sections 4.3 and 4.4.
   \end{itemize}
~\\
\item{\bf Week 4.}
   \begin{itemize}
   \item{\it Mon., July 15:}
      Sections 4.5 and review.
      {\bf Homework due:} Odd numbered problems from 4.1--4.4.
   \item{\it Tue., July 16:}
   {\bf Exam 3: Chapter 4.}
   \item{\it Wed., July 17:}
   Ch 5. Radicals and Complex Numbers
   Sections 5.1 and 5.2.
   \item{\it Thr., July 18:}
   Sections 5.3 and 5.4.
   \item{\it Fri., July 19:}
   Sections 5.5 and 5.6.
      {\bf Homework due:} Odd numbered problems from 5.1--5.4.
   \end{itemize}
\item{\bf Week 5.}
   \begin{itemize}
   \item{\it Mon., July 22:}
   {\bf Exam 4: Chapter 5.}
   \item{\it Tue., July 23:}
   Ch 6. Quadratic Equations and Inequalities:
   Sections 6.1 and 6.2.
   \item{\it Wed., July 24:}
   Sections 6.3 and 6.4.
   \item{\it Thr., July 25:}
   Sections 6.5 and 6.6.
      {\bf Homework due:} Odd numbered problems from 6.1--6.4.
   \item{\it Fri., July 26:}
   {\bf Exam 5: Chapter 6.}
   \end{itemize}
\item{\bf Week 6.}
   \begin{itemize}
   \item{\it Mon., July 29:}
   Ch 7. Introductions to Analytic Geometry:
   Sections 7.1 and 7.2.
   \item{\it Tue., July 30:}
   Sections 7.4 and 7.5.
      {\bf Homework due:} Odd numbered problems from 7.1 and 7.2.
   \item{\it Wed., July 31:}
   {\bf Exam 6: Chapter 7.}
   \end{itemize}
\end{itemize}



\end{itemize}

\end{document}
